Author’s response to reviews

Title: Comparison of conventional and Si-photomultiplier-based PET systems for image quality and diagnostic performance

Authors:
Jenny Oddstig (jenny.oddstig@skane.se)
Sigrid Leide Svegborn (sigrid.leide_svegborn@med.lu.se)
Helen Almquist (helen.almquist@skane.se)
Ulrika Bitzén (ulrika.bitzen@skane.se)
Sabine Garpered (sabine.garpered@skane.se)
Fredrik Hedeer (fredrik.hedeer@skane.se)
Cecilia Hindorf (cecilia.hindorf@skane.se)
Jonas Jögi (jonas.jogi@med.lu.se)
Lena Jönsson (lena_m.jonsson@med.lu.se)
David Minarik (david.minarik@med.lu.se)
Richard Petersson (Richard.petersson@med.lu.se)
Annika Welinder (annika.welinder@med.lu.se)
Per Wollmer (per.wollmer@med.lu.se)
Elin Trägårdh (elin.tragardh@med.lu.se)

Version: 1 Date: 20 Mar 2018

Author’s response to reviews:

1. In the Ethics approval and consent to participate section of the Declarations please clarify whether written or verbal consent was obtained from the participants. If verbal, please state the reason and whether the ethics committee approved this procedure.
RESPONSE: Written consent was obtained. This is clarified in the manuscript.
2. We note that you have included your figures in the manuscript and have uploaded them as separate table files. Please only provide one copy of each figure and remove the figures in the manuscript. Please note that all figure titles/legends should be placed at the end of the main manuscript under the heading ‘Figure legends’, after the References, and not within any of the figure files. RESPONSE: Figures are removed from main manuscript.

3. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
RESPONSE: The funding bodies had no role in the design, data collection, data analysis, data interpretation or writing process. This is clarified in the manuscript.

4. Please upload your revised manuscript as a clean copy and remove the manuscript with tracked changes from the additional files as it is no longer required at this stage of the editorial process.
RESPONSE: No track changes are used.